To the Editor of the World Journal of Clinical Oncology;

Manuscript 90696 - Minireview

We appreciated the commented made by the Revisors of our paper entitle “Current Interventional Options for Palliative Care for patients with Advanced Stage Cholangiocarcinoma”

We proceeded accordingly as per request:

1- Figures and tables illustrations were included.
2- The technique of endoscopic ultrasound guided Endoscopic biliary drainage was included and introduced in the text.
3- Language evaluation: The English-language grammatical presentation was reviewed and improved. The errors in grammar and format were fixed, throughout the entire manuscript.
4- The filled conflict-of-interest disclosure form is attached.
5- The PMID numbers and DOI citation numbers to the reference list and list all authors of the references were included
6- Core tip section was included. The number of words should be controlled between 50-100 words.
7- The author's contribution section was included

Thank you very much for the opportunity to present our paper again,

Sincerely yours,

Marcelo A. F. Ribeiro Jr., MD, MSc, PhD, FACS, FISS,MAMSE, ACC (Hon)
Chair – Division of Trauma, Critical Care and Acute Care Surgery SSMC
Chairman Education Committee of the Panamerican Trauma Society
Member of the Academy of Master Surgeons Educators - ACS
mjuniorsmca@ssmc.ae